Author’s response to reviews

Title: HPV positive, wild type TP53, and p16 overexpression correlate with the absence of residual tumors after chemoradiotherapy in anal squamous cell carcinoma.

Authors:

Paulo Soares (psoares19@hotmail.com)
Eliana Abdelhay (eabdelhay@inca.gov.br)
Luiz Claudio Thuler (lthuler@inca.gov.br)
Bruno Soares (soaresbm86@gmail.com)
Samia Demachki (demachki@gmail.com)
Gessica Valéria Ferro (gessica.ifpa@gmail.com)
Paulo Assumpção (assumpcaopp@gmail.com)
Letícia Lamarão (letlamarao@hotmail.com)
Luis Felipe Pinto (lfrpinto@inca.gov.br)
Rommel Burbano (rommel@ufpa.br)

Version: 2 Date: 09 Feb 2018

Author’s response to reviews:

Following the Editor's instructions, at this stage, we send our manuscript as a single, final, and clean version, with no changes, comments, highlights, scrawled or crawled text in different colors. All tables and figures are also being forwarded in clean versions.

Thank you very much